Author's response to reviews

Title: Potential risk factors of diabetic neuropathy: a case control study

Authors:

Fargol Booya (emrc@tums.sina.ac.ir)
Bagher Larijani (emrc@tums.sina.ac.ir)
Mohammad Pajouhi (emrc@tums.sina.ac.ir)
Mahdi Nooraei (emrc@tums.sina.ac.ir)
Jamshid Lotfi (emrc@tums.sina.ac.ir)
Fatemeh Bandarian (emrc@tums.sina.ac.ir)

Version: 2 Date: 24 July 2005

Author's response to reviews:

Dear Editor
As you requested a native english speaker read and edited our manuscript. The new version is submitted. If there is any problem no hesitate to contact us.
Kind Regards,
Bagher Larijani, MD